Author's response to reviews

Title: Prevalence of thyroid nodules in an occupationaly radiation exposed group: a cross sectional study in an area with mild iodine deficiency

Authors:

Paolo Trerotoli (trerotoli@igiene.uniba.it)
Anna Ciampolillo (a.ciampolillo@endo.uniba.it)
Giuseppe Marinelli (giuseppe_marinelli@libero.it)
Riccardo Giorgino (r.giorgino@endo.uniba.it)
Gabriella Serio (seriogab@igiene.uniba.it)

Version: 5  Date: 4 July 2005

Author's response to reviews:

Dear Editor, since your last e-mail (04-july-2005) I have revised the formatting problem that you have underlined. Tables are made as TABLE and not as tabular text. In this version I have bordered every line, but this way the tables are not so nice, but anyway they are divided and bordered.
Thanks for your quick service.
Paolo TRerotoli